Author's response to reviews

Title: The potential prognostic value of connexin 26 and 46 expression in neoadjuvant-treated breast cancer

Authors:

Ivett Teleki (telekiivett@gmail.com)
Tibor Krenacs (krenacst@gmail.com)
Marcell A Szasz (szaszam@gmail.com)
Janina Kulka (janinakulka@gmail.com)
Barna Wichmann (wwbarna@yahoo.com)
Cornelia Leo (Cornelia.Leo@usz.ch)
Barbel Papassotiropoulos (baerbelmaria@hotmail.com)
Cosima Riemenschnitter (cosimari@hotmail.com)
Holger Moch (holger.moch@usz.ch)
Zsuzsanna Varga (zsuzsanna.varga@usz.ch)

Version: 6 Date: 17 December 2012

Author's response to reviews: see over
Dear Doctor Linn,

Many thanks for your good news of accepting our manuscript by Teleki et al…. for publication in BMC Cancer.

We have checked and corrected the typographical errors and consolidated the language into UK English all over the manuscript. Also, we rephrased the sentence in the conclusion as you suggested.

In addition, I am also part of the recently established Tumor Progression Research Group of the Hungarian Academy of Sciences (MTA in Hungarian), which name has been most recently consolidated. This was also corrected at my affiliation of the title page.

New changes are underlined using red lines in the revised text.

Many thanks again!

Best regards:
Tibor Krenacs